Author's response to reviews

Title: The potential of Virtual Reality as anxiety management tool: a randomized controlled study in a sample of patients affected by Generalized Anxiety Disorder

Authors:

Alessandra Gorini (alessandra.gorini@gmail.com)
Giuseppe Riva (giuseppe.riva@unicatt.it)

Version: 4 Date: 13 March 2008

Author's response to reviews:

Dear Editor,

as you requested, we have specified the number of subjects included in the study, in the abstract (pag 2) and in the main text (pag 7-8).

Pag 2
The trial is based on a randomized controlled study, including three groups of 25 patients each (for a total of 75 patients): (1) the VR group, (2) the non-VR group and (3) the waiting list (WL) group.

Pag 7-8
A power analysis was conducted in order to evaluate the number of patients needed to detect a significant difference (Effect size $f \geq 0.40$) between the three groups (VR, non VR and WL) with a power of 0.90. The analysis showed a necessary total sample size of 75, 25 in each group, with $\alpha = 0.5$, two-tailed.

Best regards
Alessandra Gorini